Author’s response to reviews

Title: Prediction of breast cancer by profiling of urinary RNA metabolites using SVM-based feature selection

Authors:

Carsten Henneges (carsten.henneges@uni-tuebingen.de)
Dino Bullinger (dino.bullinger@uni-tuebingen.de)
Richard Fux (richard.fux@med.uni-tuebingen.de)
Natascha Friese (n.friese@web.de)
Harald Seeger (harald.seeger@med.uni-tuebingen.de)
Hans Neubauer (hans.neubauer@med.uni-tuebingen.de)
Stefan Laufer (stefan.laufer@uni-tuebingen.de)
Christoph H. Gleiter (christoph.gleiter@med.uni-tuebingen.de)
Matthias Schwab (matthias.schwab@ikp-stuttgart.de)
Andreas Zell (andreas.zell@uni-tuebingen.de)
Bernd Kammerer (bernd.kammerer@uni-tuebingen.de)

Version: 6  Date: 31 March 2009

Author’s response to reviews:

We finally made two minor revisions to the manuscript:

First we corrected the link of reference [1]. It was reset to the wrong internet address during reformating the bibliography using the Reference Manager. The correct address now is "http://www.who.int/entity/healthinfo/statistics/bodgbdeathdalyestimates.xls".

Second we inserted a space between "4" and "Institute of Pharmacy, Auf der Morgenstelle 8, D-72076 Tübingen, Germany" on the title page.

These changes were made to the most recent version of the manuscript available at "http://www.biomedcentral.com/author/manuscript/signoff/view.htm?manuscriptId=7225815112424323".

We apologize for this inconvenience and now consider the manuscript to be ready for final publication.

Sincerely,

Carsten Henneges